Author’s response to reviews

Title: Is perception of quality more important than technical quality in patient video cases?

Authors:

Damian Roland (dr98@le.ac.uk)
David Matheson (djmatheson@icloud.com)
Nic Taub (nat2@leicester.ac.uk)
Tim Coats (tc61@le.ac.uk)
Monica Lakhanpaul (m.lakhanpaul@ucl.ac.uk)

Version: 2
Date: 6 May 2015

Author’s response to reviews:

1) The e-mail addresses for the authors have been added.
2) Line numbers have been added
3) A sentence on ethical approval has been added.